Author’s response to reviews

Title: Computed Tomography Volumetry of esophageal cancer - The role of Semiautomatic assessment

Authors:

Yi-Hua Zhang (yi-hua.zhang@ki.se)
Michael Fischer (dr.michael.fischer@gmail.com)
Henrik Lehmann (henrik.lehmann@karolinska.se)
Åse Johnsson (ase.johnsson@vgregion.se)
Ioannis Rouvelas (Ioannis.Rouvelas@ki.se)
Gunnar Herlin (gunnar.herlin@karolinska.se)
Lars Lundell (lars.lundell@ki.se)
Torkel Brismar (torkel.brismar@gmail.com)

Version: 1 Date: 29 Nov 2017

Author’s response to reviews:

Dear Editor,

Thank you for your comments.

We have made the following revisions:

- Included a List of Abbreviations used in our manuscript
- Added email addresses for the authors on the title page
- Clarified that written informed consent was obtained from all patients participating in this study.
- Clarified the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript
- Added a statement that all authors have read and approved the manuscript

We thank you for your consideration
Best regards

Yi-Hua Zhang